Author's response to reviews

Title: Rare hereditary pattern in sisters with epidermodysplasia verruciformis: a case report

Authors:

Rui Yoshida (ruirui0125cherry@nms.ac.jp)
Toshihiko Kato (toshihiko.kato@phms.co.jp)
Masahiko Kawase (kawasem@iikei.ac.jp)
Mariko Honda (m-honda@iikei.ac.jp)
Tsuyoshi Mitsuishi (tmitsu@tymc.twmu.ac.jp)

Version: 4 Date: 3 May 2014

Author's response to reviews:

Dear Journal Editorial office,

We changed format into instructions of the Journal Editorial Office.

1. Abstract

We changed ‘Conclusions’ and the abstract was structured into three sections, such as Background, Case presentation, and Conclusions.

2. We added the acknowledgement section between the Author's contributions section and reference list.

3th May 2014

Sincerely yours,

Tsuyoshi Mitsuishi, MD, PhD
Department of Dermatology
Tokyo Women's Medical University
Yachiyo Medical Center
477-96, Ohwada-Shinden, Yachiyo,
Chiba, 276-8524, JAPAN
Phone: +81-47-450-6000
Fax: +81-47-458-7047
E-mail: tmitsu@tymc.twmu.ac.jp